Author’s response to reviews

Title: The case for a universal hepatitis C vaccine to achieve hepatitis C elimination

Authors:

Nick Scott (nick.scott@burnet.edu.au)
David Wilson (david.wilson@burnet.edu.au)
Alexander Thompson (alexander.thompson@svhm.org.au)
Ellie Barnes (ellie.barnes@ndm.ox.ac.uk)
Manal El-Sayed (manalhelsayed@yahoo.co.uk)
Adele Benzaken (adelebenz@gmail.com)
Heidi Drummer (heidi.drummer@burnet.edu.au)
Margaret Hellard (margaret.hellard@burnet.edu.au)

Version: 2 Date: 05 Aug 2019

Author’s response to reviews:

Please see attached point-by-point response document.